Author’s response to reviews

Title: Maternal Vitamin D Status, Gestational Diabetes and Infant Birth Size

Authors:

Helena Hauta-alus (helena.hauta-alus@helsinki.fi)
Heli Viljakainen (heli.viljakainen@helsinki.fi)
Elisa Holmlund-Suila (elisa.holmlund@helsinki.fi)
Maria Enlund-Cerullo (maria.enlund-cerullo@hus.fi)
Jenni Rosendahl (jenni.rosendahl@hus.fi)
Saara Valkama (saara.valkama@hus.fi)
Otto Helve (otto.helve@helsinki.fi)
Timo Hytinantti (timo.hytinantti@hus.fi)
Outi Mäkitie (outi.makitie@helsinki.fi)
Sture Andersson (sture.andersson@hus.fi)

Version: 2 Date: 27 Oct 2017

Author’s response to reviews:

Editor Comments:

Please move list of abbreviations to Declarations section and add date of trial registration.

Response: Thank you. We have done these modifications. Abbreviations are now in lines 358-359, page 19. The date of the registration was November 8, 2012, and this information was added into the manuscript (line 49, page 3; line 366, page 19).

BMC Pregnancy and Childbirth operates a policy of open peer review, which means that you will be able to see the names of the reviewers who provided the reports via the online peer review system. We encourage you to also view the reports there, via the action links on the left-hand side of the page, to see the names of the reviewers.
Reviewer reports:

Kathryn Hart (Reviewer 2): Thank you for responding so thoroughly to my previous comments. I agree these have greatly improved the manuscript. There are only a couple of minor language corrections that I would recommend:

Line 93 & 100 & 165 - 'communal' - should this be community?
Response: Thank you. We have done the suggested correction (line 91, page 5; line 98, page 6; line 163, page 8).

Line 324 - 'needs' not need
Response: Thank you. We have done the correction (line 322, page 17).

Line 326 - explanation for this (not 'to' this)
Response: We have done the correction (line 324, page 17).

Line 343 - 'A' further limitation
Response: We have done the correction (line 340, page 18).

Line 349 - in a vitamin D sufficient population
Response: We have done the correction (line 347, page 18).